Author’s response to reviews

Title: Effect of homeopathic drug, Chelidonium, in amelioration of p-DAB induced hepatocarcinogenesis in mice

Authors:

Surjyo Jyoti Biswas (suriyobiswas@yahoo.com)
Anisur Rahman Khuda-Bukhsh (arkb@klyuniv.ernet.in)

Version: 3 Date: 6 Apr 2002

From:
Dr. A.R. Khuda-Bukhsh, Ph.D.
Reader.

To
Ms Emma Veitch 06.04.02
Assistant Editor,
Biomedcentral, Middlesex House,
34-42Cleveland,
London W1T 4LB
Ref: 158424179101361: Potentized homeopathic drugs Chelidonium-30...........during hepatocarcinogenesis.

Dear Ms Emma Veitch,
I am resubmitting herewith our re-revised paper after taking care of all of your suggestions mentioned in your mail.
The re-revised paper has dealt with the following:
1. The title has been changed as per your suggestion.
2. The Abstract has been shortened to 250 words.
3. A subsection "Methods" has been added to the Abstract section as desired.
4. The language of the manuscript has been checked by an expert conversant with English and a few suggested changes have been made therein, e.g. in Background section,"to be referred to henceforth as" deleted.; In Materials and Methods section,"made up of" changed to "made of", "Each group were" changed to "Each group was"; "additionally" has been replaced with "also"; In Results and Discussion section, "more ability" has been changed to "greater efficacy; "marginally" has been changed to "slightly"; "It had earlier been.......end up with neoplastic characteristics", the tense of this sentence changed to "past tense".
5. Quality of Figures have been improved in PDF files, and now the legends are quite readable.

I hope that the paper has now met all the requirements suggested by the Editorial Board and would be finally accepted for publication at the earliest.

With kind regards,
Yours sincerely,

A.R. Khuda-Bukhsh

Dr. A.R. Khuda-Bukhsh, Ph.D.
Reader, Deptt. of Zoology,
University of Kalyani,
Kalyani-741235, W.B., India.